Author's response to reviews

Title: Three-dimensional Drip Infusion CT Cholangiography in patients with suspected obstructive biliary disease: A retrospective analysis on adverse reaction to contrast material and feasibility.

Authors:

Anders Persson (anders.persson@cmiv.liu.se)
Nils Dahlstrom (nils.dahlstrom@lg.se)
Orjan Smedby (orjan.smedby@imv.liu.se)
Torkel B Brismar (torkel.brismar@karolinska.se)

Version: 4  Date: 5 April 2006

Author’s response to reviews:

We are grateful for the comments by the reviewers.

Spelling and grammatical errors have been corrected by a native English-speaking colleague.

The background in the abstract section has been extended. Subtitles in the discussion section have been removed and the abbreviation CTC has been used throughout.

In Fig 2 a typing error has been corrected (umol/ml changed to umol/L).